Author's response to reviews

Title: Interaction between anaemia and hypertension on chronic kidney disease: a survey for patients with hypertension and diabetes mellitus in Taiwan

Authors:

Sui-Lung Su (a131419@gmail.com)
Chin Lin (xup6fup@hotmail.com)
Hsin-Yi Yang (b262728@gmail.com)
Chia-Chao Wu (wucc@mail.ndmctsgh.edu.tw)
Sen-Yeong Kao (kao@mail.ndmctsgh.edu.tw)
Hung-Yi Chiu (hychiou@tmu.edu.tw)
Kuo-Cheng Lu (kuochenglu@gmail.com)
Yu-Lung Chiu (long_ruth0624@mail.ndmctsgh.edu.tw)
Jin-Shuen Chen (dgschen@ndmctsgh.edu.tw)
Fung-Chang Sung (fcsung@mail.cmu.edu.tw)
Ying-Chin Ko (yoko@mail.cmu.edu.tw)
Chien-Te Lee (chientel@gmail.com)
Yu Yang (2219@cch.org.tw)
Chih-Wei Yang (cwyang@adm.cgmh.org.tw)
Shang-Jyh Hwang (sijhwang@kmu.edu.tw)
Ming-Cheng Wang (wangmc@mail.ncku.edu.tw)
Yung-Ho Hsu (yhhshu@tmu.edu.tw)
Mei-Yi Wu (e220121@gmail.com)
Yu-Mei Hsueh (ymhsueh@tmu.edu.tw)
Yuh-Feng Lin (linfymd@tmu.edu.tw)

Version: 2
Date: 14 October 2014

Author's response to reviews: see over
Dear Editor,

I would be grateful and appreciative if you could consider the enclosed manuscript entitled “Interaction between Anaemia and Hypertension on Chronic Kidney Disease: A Survey for Patients with Hypertension and Diabetes Mellitus in Taiwan” for publication in your Journal. All authors have contributed to, seen, and approved the final, submitted version of the manuscript. It has not been published elsewhere.

Best wishes,

Sincerely yours,

Yuh-Feng Lin, MD
Division of Nephrology, Department of Medicine, Shuang Ho Hospital, Graduate Institute of Clinical Medicine, Taipei Medical University, No.250, Wuxing St., Xinyi Dist., Taipei City 110, Taiwan, ROC.
E-mail: linyf@mail.ndmctsgh.edu.tw